% version 3 Febr. 2014
\documentclass[11pt]{amsart}
\usepackage{graphicx, color}
\usepackage{amscd}
\usepackage{amsmath}
\usepackage{amsfonts}
\usepackage{amssymb}
\usepackage{mathrsfs}

\textwidth=6in \textheight=9.5in \topmargin=-0.5cm
\oddsidemargin=0.5cm \evensidemargin=0.5cm
%\usepackage[notref,notcite]{showkeys}

\newtheorem{theorem}{Theorem}
\newtheorem{ex}{Example}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{prop}[theorem]{Proposition}
\newtheorem{remark}{Remark}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{claim}{Claim}
\newtheorem{step}{Step}
\newtheorem{case}{Case}
\newtheorem{definition}[theorem]{Definition}

%\newenvironment{proof}[1][Proof]{\textbf{#1.} }{\ \rule{0.5em}{0.5em}}
\newenvironment{proof-sketch}{\noindent{\bf Sketch of Proof}\hspace*{1em}}{\qed\bigskip}


%\numberwithin{equation}{section}
\newcommand{\RR}{\mathbb R}
\newcommand{\NN}{\mathbb N}
\newcommand{\PP}{\mathbb P}
\newcommand{\ZZ}{\mathbb Z}
\newcommand{\di}{\displaystyle}
\newcommand{\ri}{\rightarrow}
\renewcommand{\le}{\leqslant}
\renewcommand{\leq}{\leqslant}
\renewcommand{\ge}{\geqslant}
\renewcommand{\geq}{\geqslant}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\baselineskip=16pt plus 1pt minus 1pt

\begin{document}
\begin{thebibliography}{99}
\bibitem{1} {\sc S. Aizicovici, N.S. Papageorgiu, V. Staicu}, {\em Degree Theory for Operators of Monotone Type and Nonlinear Elliptic Equations with Inequality Constraints}, Memoirs Amer. Math. Soc., November 2008, Volume \textbf{196}, No. \textbf{915}.

\bibitem{2} {\sc A. Ambrosetti, H. Brezis, G. Cerami},  Combined effects of concave and convex nonlinearities in some elliptic problems, {\it J. Functional Analysis} \textbf{122} (1994), 519-543.

\bibitem{ambrab} {\sc A. Ambrosetti, P.H. Rabinowitz},
Dual variational methods in critical point theory and applications, {\it
J. Functional Analysis} {\bf 14} (1973), 349-381.

\bibitem{3} {\sc H. Brezis, L. Nirenberg},  $H^1$ versus $C^1$ local minimizers, {\it  C.~R. Acad. Sci. Paris} \textbf{317} (1993), 465-472.

\bibitem{4} {\sc E. O. de Paiva, E. Massa},  Multiple solutions for some elliptic equations with a nonlinearity concave at the origin, {\it  Nonlinear Analysis} \textbf{66} (2007), 2940-2946.

\bibitem{5} {\sc L. Gasinski, N.S. Papageorgiou}, Pairs of nontrivial solutions for resonant Neumann problems, {\it  J.~Math. Anal. Appl.} {\bf 398} (2013), 649-663.

\bibitem{6} {\sc D. Motreanu, N.S. Papageorgiou},  Existence and multiplicity of solutions for Neumann problems, {\it  J. Differential Equations}  \textbf{232} (2007), 1-35.

\bibitem{7} {\sc D. Motreanu, N.S. Papageorgiou},  Multiple solutions for nonlinear Neumann problems driven by a nonhomogeneous differential operator, {\it  Proc. Amer. Math. Soc.} \textbf{139} (2011), 3527-3535.

\bibitem{8} {\sc D. Mugnai, N.S. Papageorgiou},  Resonant nonlinear Neumann problems with indefinite weight, {\it  Ann. Sc. Norm. Super. Pisa Cl. Sci.} \textbf{5} (2012), 729-778.

\bibitem{9} {\sc N.S. Papageorgiou, S.T. Kyritsi}, {\em Handbook of Applied Analysis}, Springer, New York (2009).

\bibitem{10} {\sc N.S. Papageorgiou, V.D. R\u adulescu}, {\em Semilinear Neumann problems with indefinite and unbounded potential and crossing nonlinearity}, Contemporary Mathematics, Vol. \textbf{595} (2013), Amer. Math. Soc., pp.~293-315.

\bibitem{11} {\sc N.S. Papageorgiou, V.D. R\u adulescu},  Coercive and noncoercive nonlinear Neumann problems with an indefinite potential, submitted.

\bibitem{12} {\sc K. Perera},  Multiplicity results for some elliptic problems with concave nonlinearities, {\it  J. Differential Equations} \textbf{140} (1997), 133-141.

\bibitem{13} {\sc P. Pucci, J. Serrin}, {\em The Maximum Principle}, Birkh\"{a}user Verlag, Basel (2007).

\bibitem{14} {\sc X-J. Wang},  Neumann problems of semilinear elliptic equations involving critical Sobolev exponents, {\it  J. Differential Equations} \textbf{93} (1991), 283-310.

\end{thebibliography}

%\hfill\today\bigskip

\title[Neumann problems with indefinite unbounded potential and concave terms]{Neumann problems with  indefinite and unbounded potential and concave terms}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\author[N.S. Papageorgiou]{Nikolaos S. Papageorgiou}
\address{National Technical University, Department of Mathematics,
				Zografou Campus, Athens 15780, Greece}
\email{\tt npapg@math.ntua.gr}

\author[V. R\u{a}dulescu]{Vicen\c{t}iu D. R\u{a}dulescu}
\address{Department of Mathematics, Faculty of Science, King Abdulaziz University,
Jeddah, Saudi Arabia}
\email{\tt vicentiu.radulescu@math.cnrs.fr}

\keywords{Concave nonlinearity, indefinite and unbounded potential, resonance, Harnack inequality, regularity theory, local minimizers.\\
\phantom{aa} 2010 AMS Subject Classification:
Primary 35J20; Secondary 35J60, 58E05}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\begin{abstract}
We consider a semilinear parametric Neumann problem driven by the negative Laplacian plus an indefinite and unbounded potential. The reaction is asymptotically linear and exhibits a negative concave term near the origin. Using variational methods together with truncation and perturbation techniques and critical groups, we show that for all small values of the parameter the problem has at least five nontrivial solutions, four of which have constant sign.
\end{abstract}

\maketitle

%\tableofcontents

\section{Introduction}\label{sec1}

Let $\Omega\subseteq \RR^{N}$ be a bounded domain with a $C^2$-boundary $\partial\Omega$. In this paper, we study the following parametric Neumann problem:
$$
\begin{array}{cl}%
\di -\Delta u(z)+\beta(z)u(z)=f(z,u(z))-\lambda |u(z)|^{q-2}u(z)\ \mbox{in}\ \Omega,\qquad \frac{\partial u}{\partial n}=0\ \mbox{on}\ \partial\Omega.
\end{array}\eqno(P_{\lambda})
$$

 In the boundary condition $n(\cdot)$ denotes the outward unit normal on $\partial\Omega$. Also, in $(P_{\lambda})$ the potential function $\beta(\cdot)$ need not be bounded and may change sign (indefinite potential). So, the differential operator $u\mapsto -\Delta u+\beta u$ is not in general coercive. In the reaction (the right-hand side of problem $(P_{\lambda})$), the function $f(z,x)$ is a Carath\'eodory function (that is, for every $x\in\RR, z\mapsto f(z,x)$ is measurable and for a.a. $z\in\Omega, x\mapsto f(z,x)$ is continuous). We assume that for a.a. $z\in\Omega$, $f(z,\cdot)$ exhibits linear growth near $\pm\infty$ and interacts with the principal eigenvalue $\hat{\lambda}_1(\beta)$ of the differential operator $u\mapsto -\Delta u+\beta u$ with Neumann boundary condition. In the other part of the reaction, namely the term $-\lambda|x|^{q-2}x$, $\lambda>0$ is a parameter and $q\in (1,2)$. So, this is a concave term which enters in the reaction with a negative sign. Hence we are dealing with an equation resonant at $\pm\infty$ with respect to $\hat{\lambda}_1(\beta)$.

 Semilinear elliptic equations with concave contributions in the reaction, were first considered by Ambrosetti, Brezis and Cerami \cite{2}, who studied problems in which the concave term enters with a positive sign and the perturbation $f(z,x)=f(x)$ is superlinear (problems with concave-convex nonlinearities). This leads to a different geometry for the problem. Problems with a negative concave term in the reaction, were examined by Perera \cite{12} and de Paiva and Massa \cite{4}. Both deal with Dirichlet equations with no potential term $\beta u$ (that is, $\beta \equiv 0$) and no resonance is permitted.

Recently the authors, see Papageorgiou and R\u adulescu \cite{10}, examined nonparametric semilinear Neumann problems with an indefinite and unbounded potential, with no concave terms and with a crossing nonlinearity. Analogous problems driven by the $p$-Laplacian, were studied by Mugnai and Papageorgiou \cite{8} and Papageorgiou and R\u adulescu \cite{11}.

\section{Mathematical Background}\label{sec2}

 Let $X$ be a Banach space and $X^{\ast}$ its topological dual. By $\langle \cdot,\cdot \rangle$ we denote the duality brackets for the pair $(X^{\ast},X)$. Let $\varphi\in C^{1}(X)$. We say that $\varphi$ satisfies the ``Palais-Smale condition" (``PS-condition" for short), if the following property holds:

``Every sequence $\{u_n\}_{n\geq 1}\subseteq X$ such that $\{\varphi(u_n)\}_{n\geq 1}\subseteq \RR\ \mbox{is bounded and}\ \varphi'(u_n)\rightarrow 0\ \mbox{in}\ X^{\ast}\ \mbox{as}\ n\rightarrow\infty,$ admits a strongly convergent subsequence".

This compactness type condition on $\varphi$, compansates for the fact that the ambient space $X$ need not be locally compact (when $X$ is infinite dimensional) and is the main tool in proving a deformation theorem. This deformation theorem leads to a minimax theory for the critical values of $\varphi$. One of the main results in that theory is the so-called {\it mountain pass} theorem, see Ambrosetti and Rabinowitz \cite{ambrab}.


In our analysis of problem $(P_{\lambda})$ in addition to the Sobolev space $H^1(\Omega)$, we will also use the Banach space $C^1(\overline{\Omega})$. This is an ordered Banach space with positive cone given by
$$C_{+}=\{u\in C^1(\overline{\Omega}): u(z)\geq 0\ \mbox{for\ all}\ z\in\overline{\Omega}\}.$$
This cone has a nonempty interior given by
$$\mbox{int}\ C_{+}=\{u\in C_{+}: u(z)>0\ \mbox{for\ all}\ z\in\overline{\Omega}\}.$$

Consider a Carath\'eodory function $f_0:\Omega\times\RR\to\RR$ which satisfies subcritical growth with respect to the $x\in\RR$ variable, that is,
$$|f_0(z,x)|\leq a(z)\left(a+|x|^{r-1}\right)\ \mbox{for a.a.}\ z\in\Omega,\ \mbox{all}\ x\in\RR,$$
with $a\in L^s(\Omega)$, $s>\frac{N}{2}$ if $N=3$, $s>1$ if $N=2$ and $s=1$ if $N=1$ and $r\in (1,2^*)$, where $2^*$ is the critical Sobolev exponent, namely $$2^*=\left\{  \begin{array}{ll}
  \frac{2N}{N-2} &\mbox{if}\ N>2 \\
    +\infty &\mbox{if}\ N=1,2.
    \end{array}\right.$$
Let $F_0(z,x)=\displaystyle\int_0^x f_0(z,\tau)d\tau$ and consider the $C^1$-functional $\varphi_0:H^1(\Omega)\to\RR$ defined by
$$\varphi_0(u)=\frac{1}{2}\|Du\|_2^2-\displaystyle\int_{\Omega}F_0(z,u(z))dz\ \mbox{for all}\ u\in H^1(\Omega).$$

The next result is essentially a particular case of Theorem $2.1$ of Motreanu and Papageorgiou \cite{7} and its proof uses the regularity result of Wang \cite{14}.
\begin{prop}\label{p2}
Assume that $u_0\in H^1(\Omega)$ is a local $C^1(\overline{\Omega})$-minimizer of $\varphi_0$, that is, there exists $\varrho_0>0$ such that
$$\varphi_0(u_0)\leq \varphi_0(u_0+h)\ \mbox{for all}\ h\in C^1(\overline{\Omega})\ \mbox{with}\ \|h\|_{C^1(\overline{\Omega})}\leq\varrho_0.$$
Then $u_0\in C^1(\overline{\Omega})$ and it is a local $H^1(\Omega)$-minimizer of $\varphi_0$, that is, there exists $\varrho_1>0$ such that
$$\varphi_0(u_0)\leq \varphi_0(u_0+h)\ \mbox{for all}\ u\in H^1(\Omega),\ \mbox{with}\ \|u\|\leq\varrho_1.$$
\end{prop}
\begin{remark}
Here and in the sequel, by $\|\cdot\|$ we denote the norm of the Sobolev space $H^1(\Omega)$. Therefore
$$\|u\|=\left(\|u\|_2^2+\|Du\|_2^2\right)^{1/2}\ \mbox{for\ all}\ u\in H^1(\Omega).$$
\end{remark}
Also, we should mention that the first such result relating local minimizers, was proved by Brezis and Nirenberg \cite{3} for functionals defined on $H_0^1(\Omega)$.

In what follows by $\tau:H^1(\Omega)\to\RR$ we denote the $C^1$-functional defined by
$$\tau(u)=\|Du\|_2^2+\displaystyle\int_{\Omega}\beta(z)u(z)^2dz\ \mbox{for all}\ u\in H^1(\Omega).$$
We assume that $\beta\in L^s(\Omega)$ with $s>\frac{N}{2}$ and also assume that $N\geq 3$, since for $N=1,\,2$, the situation is straightforward due to the Sobolev embedding theorem. Let $s'$ denote the conjugate exponent of $s$ (that is, $\frac{1}{s}+\frac{1}{s'}=1$). We have $s'<\frac{N}{N-2}$ and so $2s'<2^*$. Then the Sobolev embedding theorem implies that $u^2\in L^{s'}(\Omega)$. Using the H\"{o}lder inequality, we have
\begin{eqnarray}\label{eq1}
\left|\displaystyle\int_{\Omega}\beta(z)u^2dz\right|\leq\|\beta\|_s\left(\displaystyle\int_{\Omega}u^{2s'}dz\right)^{\frac{1}{s'}}=\|\beta\|_s\|u\|_{2s'}^2.
\end{eqnarray}
We know that
$$H^1(\Omega)\hookrightarrow L^{2s'}(\Omega)\hookrightarrow L^2(\Omega)$$
and by the Sobolev embedding theorem, the first embedding is compact. So, invoking Ehrling's inequality (see, for example, Papageorgiou and Kyritsi \cite[p. 698]{9}), given $\varepsilon>0$, we can find $c(\varepsilon)>0$ such that
\begin{eqnarray}\label{eq2}
\|u\|_{2s'}^2\leq\varepsilon\|u\|^2+c(\varepsilon)\|u\|_2^2\quad \mbox{for all}\ u\in H^1(\Omega).
\end{eqnarray}
From (\ref{eq1}) and (\ref{eq2}), we see that for all $u\in H^1(\Omega)$, we have
\begin{eqnarray}\label{eq3}
&&\|Du\|_2^2+\|u\|_2^2-\displaystyle\int_{\Omega}\beta(z)u^2dz\leq\|Du\|_2^2+\|u\|_2^2+\varepsilon\|\beta\|_s\|u\|^2+c(\varepsilon)\|\beta\|_s\|u\|_2^2 \nonumber\\
&\Rightarrow& (1-\varepsilon\|\beta\|_s)\|u\|^2\leq\tau(u)+(1+c(\varepsilon)\|\beta\|_s)\|u\|_2^2.
\end{eqnarray}
If we choose $\varepsilon\in\left(0,\frac{1}{\|\beta\|_s}\right)$, then from (\ref{eq3}) it follows that
\begin{eqnarray}\label{eq4}
\hat{c}\|u\|^2\leq\tau(u)+\gamma\|u\|_2^2\ \mbox{for all}\ u\in H^1(\Omega),\ \mbox{with}\ \hat{c}\in (0,1),\ \gamma>0.
\end{eqnarray}
We consider the following linear eigenvalue problem
\begin{eqnarray}\label{eq5}
-\Delta u(z)+\beta(z)u(z)=\hat{\lambda}u(z)\ \mbox{in}\ \Omega,\qquad \frac{\partial u}{\partial n}=0\ \mbox{on}\ \partial\Omega.
\end{eqnarray}
From Papageorgiou and R\u adulescu \cite{10}, we know that problem (\ref{eq5}) admits a sequence $\{\hat{\lambda}_k(\beta)\}_{k\geq 1}$ of eigenvalues such that  $\hat{\lambda}_k(\beta)\to +\infty$ as $k\to \infty$ and the corresponding eigenfunctions belong in $C^1(\overline{\Omega})$ and form orthogonal bases for the separable Hilbert spaces $L^2(\Omega)$ and $H^1(\Omega)$. The principal eigenvalue $\hat{\lambda}_1(\beta)$ is simple (that is, the corresponding eigenspace is one dimensional) and admits the following variational characterization:
\begin{eqnarray}\label{eq6}
\hat{\lambda}_1(\beta)=\inf\left[\frac{\tau(u)}{\|u\|_2^2}: u\in H^1(\Omega),u\neq 0\right].
\end{eqnarray}
The infimum is realized on the corresponding one-dimensional eigenspace. Let $\hat{u}_1(\beta)\in C^1(\overline{\Omega})$ be the $L^2$-normalized (that is, $\|\hat{u}_1(\beta)\|_2=1$) principal eigenfunction. From (\ref{eq6}) we see that it does not change sign and we choose it to be positive, that is, $\hat{u}_1(\beta)\in C_{+}$. Moreover, from Harnack's inequality (see Pucci and Serrin \cite[p. 163]{13}), we obtain $\hat{u}_1(\beta)(z)>0$ for all $z\in\Omega$. In fact if we assume that $\beta^{+}\in L^{\infty}(\Omega)$, using the maximum principle (see Pucci and Serrin \cite[p. 120]{13}), we can conclude that $\hat{u}_1(\beta)\in$ int $C_{+}$. We mention that $\hat{\lambda}_1(\beta)$ is the only eigenvalue with eigenfunctions of constant sign. All the other eigenvalues have nodal (sign changing) eigenfunctions. By $E(\hat{\lambda}_k(\beta))$ we denote the eigenspace corresponding to the eigenvalue $\hat{\lambda}_k(\beta)$. We have the following orthogonal direct sum decomposition
$$H^1(\Omega)=\overline{\oplus_{k\geq 1}E(\hat{\lambda}_k(\beta))}.$$
The nonprincipal eigenvalues have the following variational characterizations:
\begin{eqnarray*}
\hat{\lambda}_m(\beta)&&=\inf\left[\frac{\tau(u)}{\|u\|^2_2}: u\in \overline{\oplus_{k\geq m}E(\hat{\lambda}_k(\beta))}\right]\\
&&=\sup\left[\frac{\tau(u)}{\|u\|^2_2}: u\in \oplus_{k=1}^m E(\hat{\lambda}_k(\beta))\right],\ m\geq 2.
\end{eqnarray*}
Both the infimum and the supremum are realized on $E(\hat{\lambda}_m(\beta))$.

Finally, we mention that the eigenspaces $E(\hat{\lambda}_k(\beta))$, $k\geq 1$, have the unique continuation property, namely if $u\in E(\hat{\lambda}_k(\beta))$ and $u$ vanishes on a set of positive Lebesgue measure, then $u\equiv 0$.

The following proposition is an easy consequence of the spectral properties of $-\Delta u+\beta(z)u$ described above. See also Papageorgiou and R\u adulescu \cite{10} (Proposition $2.3$ for $(a)$ and Gasinski and Papageorgiou \cite{5} for $(b)$, $(c)$).
\begin{prop}\label{p3}
\begin{itemize}
\item[(a)] If $\vartheta\in L^{\infty}(\Omega)$, $\vartheta(z)\leq \hat{\lambda}_1(\beta)$ a.e. in $\Omega$, $\vartheta\neq \hat{\lambda}_1(\beta)$, then there exists $c_0>0$ such that
    $$\tau(u)-\displaystyle\int_{\Omega}\vartheta(z)u^2dz\geq c_0\|u\|^2\ \mbox{for all}\ u\in H^1(\Omega).$$
\item[(b)] If $\eta\in L^{\infty}(\Omega)$, $\eta(z)\leq \hat{\lambda}_{m+1}(\beta)$ a.e. in $\Omega$, $\eta\neq \hat{\lambda}_{m+1}(\beta)$, then there exists $\tilde{\xi}_0>0$ such that
 $$\tau(u)-\displaystyle\int_{\Omega}\eta(z)u^2dz\geq \tilde{\xi_0}\|u\|^2\ \mbox{for all}\ u\in \overline{\oplus_{k\geq m+1}E(\hat{\lambda}_k(\beta))}.$$
\item[(c)] If $\eta\in L^{\infty}(\Omega)$, $\eta(z)\geq \hat{\lambda}_{m}(\beta)$ a.e. in $\Omega$, $\eta\neq \hat{\lambda}_{m}(\beta)$, then there exists $\hat{\xi}_0>0$ such that
 $$\tau(u)-\displaystyle\int_{\Omega}\eta(z)u^2dz\leq -\hat{\xi}_0\|u\|^2\ \mbox{for all}\ u\in \oplus_{k=1}^m E(\hat{\lambda}_k(\beta)).$$
\end{itemize}
\end{prop}
Let $(Y_1,Y_2)$ be a topological pair such that $Y_2\subseteq Y_1\subseteq X$. For every integer $k\geq 0$ we denote by $H_k(Y_1,Y_2)$ the $k$th relative singular homology group with integer coefficients for the pair $(Y_1,Y_2)$.

Given $\varphi\in C^1(X)$ and $c\in\RR$, we introduce the following sets:
$$\varphi^c=\{u\in X: \varphi(u)\leq c\},\ K_{\varphi}=\{u\in X: \varphi'(u)=0\},\ K_{\varphi}^c=\{u\in K_{\varphi}: \varphi(u)=c\}.$$

Let $u\in X$ be an isolated critical point of $\varphi$ with $\varphi(u)=c$ (that is, $u\in K_{\varphi}^c$). Then the critical groups of $\varphi$ at $u$ are defined by
$$C_k(\varphi,u)=H_k(\varphi^c\cap U,\varphi^c\cap U\setminus\{u\})\ \mbox{for all}\ k\geq 0,$$
where $U$ is a neighborhood of $u$ such that  $K_{\varphi}\cap\varphi^c\cap U=\{u\}$. The excision property of the singular homology theory implies that the above definition of critical groups is independent of the particular choice of the neighborhood $U$.

Finally we mention that if $h:\Omega\times\RR\ri\RR$ is a measurable function (for example, a Carath\'eodory function), we denote by $N_h$ the Nemytski operator corresponding to $h$, that is,
$$N_h(u)(x)=h(\cdot ,u(\cdot))\ \ \mbox{for all $u\in H^1(\Omega)$}.$$

\section{Solutions of Constant Sign}\label{sec3}
In this section, we produce solutions of constant sign for problem $(P_{\lambda})$ when $\lambda>0$ is sufficiently small. To do this, we introduce the following hypotheses on the data of the problem.

\smallskip
$H_0$: $\beta\in L^s(\Omega)$ with $s>\frac{N}{2}$.

$H_1$: $f:\Omega\times\RR\to\RR$ is a Carath\'eodory function such that  $f(z,0)=0$ for a.a. $z\in\Omega$ and
\begin{itemize}
\item [(i)] $|f(z,x)|\leq a(z)(1+|x|)$ for a.a. $z\in\Omega$, all $x\in\RR$ with $a\in L^{\infty}(\Omega)_{+}$;
\item [(ii)] $\displaystyle\limsup_{x\to\pm\infty}\frac{f(z,x)}{x}\leq\hat{\lambda}_1(\beta)$ uniformly for a.a. $z\in\Omega$;
\item [(iii)] $\displaystyle\lim_{x\to\pm\infty}[f(z,x)x-2F(z,x)]=+\infty$ uniformly for a.a. $z\in\Omega$, where $F(z,x)=\displaystyle\int_0^x f(z,\tau)d\tau$;
\item [(iv)] there exist functions $\eta_0,\hat{\eta}_0\in L^{\infty}(\Omega)$ such that
\begin{eqnarray*}
&&\hat{\lambda}_1(\beta)\leq \eta_0(z)\ \mbox{a.e. in}\ \Omega,\ \eta_0\neq\hat{\lambda}_1(\beta)\\
&&\eta_0(z)\leq \displaystyle\liminf_{x\to 0}\frac{f(z,x)}{x}\leq\displaystyle\limsup_{x\to 0}\frac{f(z,x)}{x}\leq\hat{\eta}_0(z)\ \mbox{uniformly for a.a.}\ z\in\Omega.
\end{eqnarray*}
\end{itemize}
\begin{remark}
Hypothesis $H_1(ii)$ implies that asymptotically at $\pm\infty$ we can have resonance with respect to the principal eigenvalue $\hat{\lambda}_1(\beta)$. This justifies the presence of hypothesis $H_1(iii)$ which is needed in order for the energy functional to satisfy the $PS$-condition. Hypothesis $H_1(iv)$ implies that at the origin we have nonuniform nonresonance with respect to $\hat{\lambda}_1(\beta)$.
\end{remark}

To produce solutions of constant sign, we introduce the following truncations-perturbations of the reaction of problem $(P_{\lambda})$:
\begin{eqnarray}\label{eq7}
&&\hat{h}_{\lambda}^{+}(z,x)=\left\{  \begin{array}{ll}
  0 &\mbox{if}\ x\leq 0 \\
  f(z,x)+\gamma x-\lambda x^{q-1} &\mbox{if}\ x>0
    \end{array}\right. \\
&&\hat{h}_{\lambda}^{-}(z,x)=\left\{  \begin{array}{ll}
  f(z,x)+\gamma x-\lambda |x|^{q-2}x &\mbox{if}\ x<0 \\
  0 &\mbox{if}\ x\geq 0.\nonumber
    \end{array}\right.
\end{eqnarray}
Here $\gamma>0$ is as in inequality (\ref{eq4}). Both $\hat{h}_{\lambda}^{\pm}(z,x)$ are Carath\'eodory functions. We set $\hat{H}_{\lambda}^{\pm}(z,x)=\displaystyle\int_0^x \hat{h}_{\lambda}^{\pm}(z,\tau)d\tau$ and consider the $C^1$-functionals $\hat{\varphi}_{\lambda}^{\pm}:H^1(\Omega)\to\RR$ defined by
$$\hat{\varphi}_{\lambda}^{\pm}(u)=\frac{1}{2}\tau(u)+\frac{\gamma}{2}\|u\|_2^2-\displaystyle\int_{\Omega}\hat{H}_{\lambda}^{\pm}(z,u(z))dz\ \mbox{for\ all}\ u\in H^1(\Omega).$$
Also let $\varphi_{\lambda}:H^1(\Omega)\to\RR$ be the energy functional for problem $(P_{\lambda})$ defined by
$$\varphi_{\lambda}(u)=\frac{1}{2}\tau(u)+\frac{\lambda}{q}\|u\|_q^q-\displaystyle\int_{\Omega}F(z,u(z))dz\ \mbox{for\ all}\ u\in H^1(\Omega).$$
Evidently $\varphi_{\lambda}\in C^1(H^1(\Omega))$.
\begin{prop}\label{p4}
If hypotheses $H_0$ and $H_1$ hold and $\lambda>0$, then the functionals $\hat{\varphi}_{\lambda}^{\pm}$ and $\varphi_{\lambda}$ are all coercive.
\end{prop}
\begin{proof}
We do the proof for $\varphi_{\lambda}^{+}$, the proofs for $\varphi_{\lambda}^{-}$ and $\varphi_{\lambda}$ being similar.

We argue indirectly. So, suppose that the functional $\varphi_{\lambda}^{+}$ is not coercive. Then we can find $\{u_n\}_{n\geq 1}\subseteq H^1(\Omega)$ and $M_1>0$ such that
$$\|u_n\|\to\infty\ \mbox{and}\ \hat{\varphi}_{\lambda}^{+}(u_n)\leq M_1\ \mbox{for\ all}\ n\geq 1.$$
So, we have
\begin{eqnarray}\label{eq8}
\frac{1}{2}\tau(u_n)+\frac{\gamma}{2}\|u_n\|_2^2-\displaystyle\int_{\Omega}\hat{H}_{\lambda}^{+}(z,u_n)dz\leq M_1\ \mbox{for\ all}\ n\geq 1.
\end{eqnarray}
Let $y_n=\frac{u_n}{\|u_n\|}$, $n\geq 1$. Then $\|y_n\|=1$ for all $n\geq 1$ and so we may assume that
\begin{eqnarray}\label{eq9}
y_n\overset{w}{\rightarrow} y\ \mbox{in}\ H^{1}(\Omega)\ \mbox{and}\ y_n\rightarrow y\ \mbox{in}\ L^2(\Omega).
\end{eqnarray}
From (\ref{eq7}) and (\ref{eq8}) we have
\begin{eqnarray}\label{eq10}
\frac{1}{2}\tau(y_n)+\frac{\gamma}{2}\|y_n^{-}\|_2^2-\displaystyle\int_{\Omega}\frac{F(z,u_n^{+})}{\|u_n\|^2}dz+\frac{\lambda}{q}\frac{1}{\|u_n\|^{2-q}} \|y_n^{+}\|_q^q\leq \frac{M_1}{\|u_n\|^2}\ \mbox{for all}\ n\geq 1.
\end{eqnarray}
By virtue of hypothesis $H_1(i)$ we have that
$$\left\{\frac{F(\cdot,u_n^{+}(\cdot))}{\|u_n\|^2}\right\}_{n\geq 1}\subseteq L^1(\Omega)\ \mbox{is uniformly integrable}.$$
So, by the Dunford-Pettis theorem and hypothesis $H_1(ii)$, we have (at least for a subsequence) that
\begin{eqnarray}\label{eq11}
\frac{F(\cdot,u_n^{+}(\cdot))}{\|u_n\|^2}\overset{w}{\rightarrow}\frac{1}{2}\vartheta(y^{+})^2\ \mbox{in}\ L^1(\Omega),
\end{eqnarray}
with $\vartheta\in L^{\infty}(\Omega)$ satisfying $\vartheta(z)\leq \hat{\lambda}_1(\beta)$ a.e. in $\Omega$ (see also Aizicovici, Papageorgiou and Staicu \cite{1}, proof of Proposition 14). Note that the functional $\tau(\cdot)$ is sequentially weakly lower semicontinuous (an easy consequence of the Sobolev embedding theorem). So, if in (\ref{eq10}) we pass to the limit as $n\to\infty$ and use (\ref{eq9}), (\ref{eq11}) and the fact that $q<2$, we obtain
\begin{eqnarray}\label{eq12}
&&\frac{1}{2}\tau(y)+\frac{\gamma}{2}\|y^{-}\|_2^2\leq \frac{1}{2}\displaystyle\int_{\Omega}\vartheta(z)(y^{+})^2dz,
\end{eqnarray}
\begin{eqnarray}\label{eq13}
&\Rightarrow& \tau(y^{+})\leq \displaystyle\int_{\Omega}\vartheta(z)(y^{+})^2dz\ (\mbox{see}\ (\ref{eq4})).
\end{eqnarray}
If $\vartheta\neq \hat{\lambda}_1(\beta)$, then from (\ref{eq13}) and Proposition \ref{p3}, we have
$$c_0\|y^{+}\|^2\leq 0,\ \mbox{hence}\ y^{+}=0.$$
Also from (\ref{eq12}) and (\ref{eq4}), we have
$$\hat{c}\|y^{-}\|^2\leq 0,\ \mbox{hence}\ y^{-}=0.$$
Therefore $y=0$ and so we have
\begin{eqnarray*}
&&\frac{\hat{H}_{\lambda}^{+}(\cdot,u_n(\cdot))}{\|u_n\|^2}\overset{w}{\rightarrow} 0 \ \mbox{in}\ L^1(\Omega)\ (\mbox{as above}),\\
&\Rightarrow& \tau(u_n)+\gamma\|u_n\|_2^2\to 0\ (\mbox{see}\ (\ref{eq8})),\\
&\Rightarrow& y_n\to 0\ \mbox{in}\ H^1(\Omega) \ (\mbox{see}\ (\ref{eq4})),
\end{eqnarray*}
a contradiction to the fact that $\|y_n\|=1$ for all $n\geq 1$.

Next we assume that $\vartheta(z)=\hat{\lambda}_1(\beta)$ a.e. in $\Omega$ (resonant case). From (\ref{eq13}) and (\ref{eq6}), we infer that
\begin{eqnarray*}
&&\tau(y^{+})=\hat{\lambda}_1(\beta)\|y^{+}\|_2^2\\
&\Rightarrow& y^{+}=\xi \hat{u}_1(\beta)\ \mbox{for some}\ \xi\geq 0.
\end{eqnarray*}

First assume that $\xi=0$. Then $y^{+}=0$ and from (\ref{eq12}) and (\ref{eq4}), we also have $y^{-}=0$. Hence $y=0$. Then reasoning as above, we obtain that $y_n\to 0$ in $H^1(\Omega)$, a contradiction to the fact that $\|y_n\|=1$ for all $n\geq 1$.

So, suppose that $\xi>0$. Then $y^{+}(z)>0$ for all $z\in\Omega$ and so $y(z)=y^{+}(z)>0$ for all $z\in\Omega$ and we have
\begin{eqnarray}\label{eq14}
u_n(z)=u_n^{+}(z)\to +\infty\ \mbox{for a.a.}\ z\in\Omega.
\end{eqnarray}
For a.a. $z\in\Omega$ and all $x>0$, we have
\begin{eqnarray}\label{eq15}
\frac{d}{dx}\left(\frac{F(z,x)}{x^2}\right)=\frac{f(z,x)x^2-2xF(z,x)}{x^4}=\frac{f(z,x)x-2F(z,x)}{x^3}.
\end{eqnarray}
By virtue of hypothesis $H_1(iii)$, given $\xi>0$, we can find $M_2>0$ such that
$$f(z,x)x-2F(z,x)\geq\xi\ \mbox{for a.a.}\ z\in\Omega,\ \mbox{all}\ x\geq M_2.$$
Using this in (\ref{eq15}), we obtain
\begin{eqnarray*}
&&\frac{d}{dx}\left(\frac{F(z,x)}{x^2}\right)\geq\frac{\xi}{x^2}\ \mbox{for a.a.}\ z\in\Omega,\ \mbox{all}\ x\geq M_2,\\
&\Rightarrow&\frac{F(z,u)}{u^2}-\frac{F(z,x)}{x^2}\geq -\frac{\xi}{2}\left[\frac{1}{u^2}-\frac{1}{x^2}\right]\ \mbox{for a.a.}\ z\in\Omega,\ \mbox{all}\ u\geq x\geq M_2.
\end{eqnarray*}
Passing to the limit as $u\to +\infty$ and using hypothesis $H_1(ii)$, we obtain
\begin{eqnarray*}
&&\frac{\hat{\lambda}_1(\beta)}{2}-\frac{F(z,x)}{x^2}\geq\frac{\xi}{2}\,\frac{1}{x^2}\ \mbox{for a.a.}\ z\in\Omega,\ \mbox{all}\ x\geq M_2\\
&\Rightarrow& \hat{\lambda}_1(\beta)x^2-2F(z,x)\geq\xi\ \mbox{for a.a.}\ z\in\Omega,\ \mbox{all}\ x\geq M_2.
\end{eqnarray*}
Since $\xi>0$ is arbitrary, we infer that
\begin{eqnarray}\label{eq16}
\hat{\lambda}_1(\beta)x^2-2F(z,x)\to +\infty\ \mbox{as}\ x\to +\infty\ \mbox{uniformly for a.a.}\ z\in\Omega.
\end{eqnarray}
From (\ref{eq8}), (\ref{eq7}) and (\ref{eq6}), we have
\begin{eqnarray}\label{eq17}
0\leq \frac{\gamma}{2}\|u_n^{-}\|_2^2\leq M_1-\displaystyle\int_{\Omega}[\hat{\lambda}_1(\beta)(u_n^{+})^2-2F(z,u_n^{+})]dz.
\end{eqnarray}
From (\ref{eq14}), (\ref{eq16}) and Fatou's lemma, it follows that
\begin{eqnarray}\label{eq18}
\displaystyle\int_{\Omega}[\hat{\lambda}_1(\beta)(u_n^{+})^2-2F(z,u_n^{+})]dz\to +\infty.
\end{eqnarray}
Comparing (\ref{eq17}) and (\ref{eq18}), we reach a contradiction. This proves the coercivity of $\hat{\varphi}_{\lambda}^{+}$. In a similar fashion, we show the coercivity of $\hat{\varphi}_{\lambda}^{-}$ and of $\varphi_{\lambda}$.
\end{proof}

From the above proposition, it follows easily the following result concerning the compactness properties of the functionals $\hat{\varphi}_{\lambda}^{\pm}$ and $\varphi_{\lambda}$.
\begin{corollary}\label{c5}
If hypotheses $H_0$ and $H_1$ hold and $\lambda>0$, then the functionals $\hat{\varphi}_{\lambda}^{\pm}$ and $\varphi_{\lambda}$ satisfy the $PS$-condition..
\end{corollary}
Proposition \ref{p4} permits the use of the direct method on the functionals $\hat{\varphi}_{\lambda}^{\pm}$. The next result guarantees that for small $\lambda>0$ the minimizers of $\hat{\varphi}_{\lambda}^{\pm}$ obtained by the direct method, are nontrivial.
\begin{prop}\label{p6}
If hypotheses $H_0$ and $H_1$ hold, then we can find $\lambda^*>0$ such that  for every $\lambda\in (0,\lambda^*)$, there exists $t^*=t^*(\lambda)>0$ for which we have
$$\varphi_{\lambda}(\pm t^*\hat{u}_1(\beta))<0.$$
\end{prop}
\begin{proof}
By virtue of hypotheses $H_1(i),(iv)$, given $\varepsilon>0$, we can find $c_1=c_1(\varepsilon)>0$ such that
\begin{eqnarray}\label{eq19}
F(z,x)\geq\frac{1}{2}(\eta_0(z)-\varepsilon)x^2-c_1|x|^r\ \mbox{for a.a.}\ z\in\Omega,\ \mbox{all}\ x\in\RR,\ \mbox{with}\ r>2.
\end{eqnarray}
For $t\neq 0$, we have
\begin{eqnarray*}
& &\varphi_{\lambda}(t\hat{u}_1(\beta))=\frac{t^2}{2}\tau(\hat{u}_1(\beta))+\frac{|t|^q\lambda}{q}\|\hat{u}_1(\beta))\|_q^q-\displaystyle\int_{\Omega} F(z,t\hat{u}_1(\beta))dz\leq \\
& & \frac{t^2}{2}\left[\tau(\hat{u}_1(\beta))-\displaystyle\int_{\Omega}(\eta_0(z)-\varepsilon)\hat{u}_1(\beta)^2dz\right]+c_1|t|^r\|\hat{u}_1(\beta)\|_r^r+ \frac{|t|^q\lambda}{q}\|\hat{u}_1(\beta)\|_q^q\ (\mbox{see}\ (\ref{eq18}))\ =\\
& &\frac{t^2}{2}\left[\displaystyle\int_{\Omega}\left(\hat{\lambda}_1(\beta)-\eta_0(z)\right)\hat{u}_1(\beta)^2dz+\varepsilon\right]+c_1|t|^r\|\hat{u}_1(\beta) \|_r^r+ \frac{t^q\lambda}{q}\|\hat{u}_1(\beta)\|_q^q\\
&&(\mbox{recall that}\ \tau(\hat{u}_1(\beta))=\hat{\lambda}_1(\beta)\|\hat{u}_1(\beta)\|_2^2\ \mbox{and}\ \|\hat{u}_1(\beta)\|_2=1).
\end{eqnarray*}
We know that $\hat{u}_1(\beta)(z)>0$ for all $z\in\Omega$. So
$$\xi^*=\displaystyle\int_{\Omega}\left(\eta_0(z)-\hat{\lambda}_1(\beta)\right)\hat{u}_1(\beta)^2dz>0\ (\mbox{see}\ H_1(iv)).$$
Choosing $\varepsilon\in (0,\xi^*)$, we obtain
\begin{eqnarray}\label{eq20}
\varphi_{\lambda}(t\hat{u}_1(\beta))&\leq&-c_2 t^2+c_3\left(|t|^r+\frac{\lambda}{q}|t|^q\right)\ \mbox{for some}\ c_2,c_3>0,\ \mbox{all}\ t\neq 0\nonumber\\
&=&\left[-c_2+c_3\left(|t|^{r-2}+\frac{\lambda}{q}|t|^{q-2}\right)\right]t^2\ \mbox{for all}\ t\neq 0.
\end{eqnarray}
Let $\mu_{\lambda}(\tau)=\tau^{r-2}+\frac{\lambda}{q}\tau^{q-2}$ for all $\tau>0$. Evidently $\mu_{\lambda}\in C^1(0,+\infty)$ and since $q<2<r$, we see that
$$\mu_{\lambda}(\tau)\to +\infty\ \mbox{as}\ \tau\to 0^{+}\ \mbox{and as}\ \tau\to +\infty.$$
So, we can find $\tau_0>0$ such that
\begin{eqnarray*}
&&\mu_{\lambda}(\tau_0)=\inf\left[\mu(\tau): \tau>0\right]>0\\
&\Rightarrow& \mu_{\lambda}'(\tau_0)=0\\
&\Rightarrow& \frac{\lambda}{q}(2-q)\tau_0^{q-3}=(r-2)\tau_0^{r-3}\\
&\Rightarrow& \tau_0=\tau_0(\lambda)=\left[\frac{\lambda (2-q)}{q(r-2)}\right]^{\frac{1}{r-q}}.
\end{eqnarray*}
Then $\mu_{\lambda}(\tau_0(\lambda))\to 0^{+}\ \mbox{as}\ \lambda\to 0^{+}$. So, we can find $\lambda^*>0$ such that
$$\mu_{\lambda}(\tau_0)<\frac{c_2}{c_3}\ \mbox{for all}\ \lambda\in (0,\lambda^*).$$
Taking $t^*=\tau_0(\lambda)$, from (\ref{eq20}) we have
$$\varphi_{\lambda}(\pm t^*\hat{u}_1(\beta))<0.$$
\end{proof}

The next proposition will be useful in applying the mountain pass theorem.
\begin{prop}\label{p7}
If hypotheses $H_0$ and $H_1$ hold and $\lambda>0$, then $u=0$ is a local minimizer for the functionals $\hat{\varphi}_{\lambda}^{\pm}$ and $\varphi_{\lambda}$.
\end{prop}
\begin{proof}
Again we do the proof for the functional $\hat{\varphi}_{\lambda}^{+}$, the proofs for the fonctionals $\hat{\varphi}_{\lambda}^{\pm}$ and $\varphi_{\lambda}$ being similar.

By virtue of hypotheses $H_1(i),(iv)$, we see that there exists $c_4>0$ such that
\begin{eqnarray}\label{eq21}
|F(z,x)|\leq c_4 x^2\ \mbox{for a.a.}\ z\in\Omega,\ \mbox{all}\ x\in\RR.
\end{eqnarray}
Then for every $u\in H^1(\Omega)$, we have
\begin{eqnarray}\label{eq22}
\hat{\varphi}_{\lambda}^{+}(u)&=&\frac{1}{2}\tau(u)+\frac{\gamma}{2}\|u\|_2^2-\displaystyle\int_{\Omega}\hat{H}_{\lambda}^{+}(z,u)dz\nonumber\\
&=&\frac{1}{2}\tau(u)+\frac{\gamma}{2}\|u^{-}\|_2^2+\frac{\lambda}{q}\|u^{+}\|_q^q-\displaystyle\int_{\Omega}F(z,u^{+})dz\ (\mbox{see}\ (\ref{eq7}))\nonumber\\
&\geq& \frac{1}{2}\tau(u^{+})+\frac{\lambda}{q}\|u^{+}\|_q^q-c_4\|u^{+}\|_2^2+\frac{1}{2}\tau(u^{-})+\frac{\gamma}{2}\|u^{-}\|_2^2\ (\mbox{see}\ (\ref{eq21}))\nonumber\\
&\geq& \frac{\hat{c}}{2}\|u^{+}\|^2-\frac{\gamma}{2}\|u^{+}\|_2^2+\frac{\lambda}{q}\|u^{+}\|_q^q-c_4\|u^{+}\|_2^2\ (\mbox{see}\ (\ref{eq4}))\nonumber\\
&\geq& \displaystyle\int_{\Omega}\left[\frac{\lambda}{q}(u^{+})^q-c_5(u^{+})^2\right]dz\ \mbox{with}\ c_5=c_4+\frac{\gamma}{2}>0.
\end{eqnarray}
Suppose that we could find $\{u_n\}_{n\geq 1}\subseteq C^1(\overline{\Omega})$ such that  $u_n\to 0$ in $C^1(\overline{\Omega})$ and
\begin{eqnarray}\label{eq23}
&&\hat{\varphi}_{\lambda}^{+}(u_n)<0,\ \mbox{hence}\ u_n^{+}\neq 0\ \mbox{for all}\ n\geq 1\ (\mbox{see}\ (\ref{eq7}))\nonumber\\
&\Rightarrow& \displaystyle\int_{\Omega}\left[\frac{\lambda}{q}-c_5\|u_n^{+}\|_{\infty}^{2-q}\right](u_n^{+})^qdz<0\ \mbox{for all}\ n\geq 1\ (\mbox{see}\ (\ref{eq22})).
\end{eqnarray}
Since $u_n\to 0$ in $C^1(\overline{\Omega})$, it follows that $u_n^{+}\to 0$ in $C(\overline{\Omega})$. So, we can find $n_0\in\NN$ such that
\begin{eqnarray}\label{eq24}
\displaystyle\int_{\Omega}\left[\frac{\lambda}{q}-c_5\|u_n^{+}\|_{\infty}^{2-q}\right](u_n^{+})^qdz>0\ \mbox{for all}\ n\geq n_0.
\end{eqnarray}
Evidently, relations (\ref{eq23}) and (\ref{eq24}) lead to a contradiction. This proves that $u=0$ is a local $C^1(\overline{\Omega})$-minimizer of $\hat{\varphi}_{\lambda}^{+}$. Invoking Proposition \ref{p2}, we conclude that $u=0$ is a local $H^1(\Omega)$-minimizer of $\hat{\varphi}_{\lambda}^{+}$.

Similarly for the functionals $\hat{\varphi}_{\lambda}^{-}$ and $\varphi_{\lambda}$.
\end{proof}

Now we are ready to produce constant sign solutions for problem $(P_{\lambda})$ when $\lambda>0$ is small, namely when $\lambda\in (0,\lambda^*)$.
\begin{prop}\label{p8}
If hypotheses $H_0$ and $H_1$ hold and $\lambda\in (0,\lambda^*)$, then problem $(P_{\lambda})$ has at least four nontrivial constant sign solutions
$$u_0,\hat{u}\in C_{+}\ \mbox{with}\ u_0(z),\hat{u}(z)>0\ \mbox{for all}\ z\in\Omega,$$
$$v_0,\hat{v}\in -C_{+}\ \mbox{with}\ v_0(z),\hat{v}(z)<0\ \mbox{for all}\ z\in\Omega.$$
Moreover, $u_0,v_0$ are local minimizers of the energy functional $\varphi_{\lambda}$ while $\hat{u},\hat{v}$ have positive energy, that is, $\varphi_{\lambda}(\hat{u}),\varphi_{\lambda}(\hat{v})>0.$
\end{prop}
\begin{proof}
By virtue of Proposition \ref{p4}, the functional $\hat{\varphi}_{\lambda}^{+}$ is coercive. Also, using the Sobolev embedding theorem, we can see that $\hat{\varphi}_{\lambda}^{+}$ is sequentially weakly lower semicontinuous. So, by the Weierstrass theorem, we can find $u_0\in H^1(\Omega)$ such that
\begin{eqnarray}\label{eq25}
\hat{\varphi}_{\lambda}^{+}(u_0)=\inf\left[\hat{\varphi}_{\lambda}^{+}(u): u\in H^1(\Omega)\right].
\end{eqnarray}
Since $\lambda\in (0,\lambda^*)$, from Proposition \ref{p6} we know that
\begin{eqnarray*}
&&\hat{\varphi}_{\lambda}^{+}(t^*\hat{u}_1(\beta))<0\\
&\Rightarrow& \hat{\varphi}_{\lambda}^{+}(u_0)<0=\hat{\varphi}_{\lambda}^{+}(0)\ (\mbox{see}\ (\ref{eq25})),\ \mbox{hence}\ u_0\neq 0.
\end{eqnarray*}
From (\ref{eq25}) we have
\begin{eqnarray}\label{eq26}
&&(\hat{\varphi}_{\lambda}^{+})'(u_0)=0\nonumber\\
&\Rightarrow& A(u_0)+(\beta(z)+\gamma)u_0=N_{\hat{h}_{\lambda}^{+}}(u_0),
\end{eqnarray}
where $A\in \mathcal{L}(H^1(\Omega),H^1(\Omega)^*)$ is defined by
$$\langle A(u),y \rangle=\displaystyle\int_{\Omega} (Du,Dy)_{\RR^{N}}dz\ \mbox{for all}\ u,y\in H^1(\Omega).$$
On (\ref{eq26}) we act with $-u_0^{-}\in H^1(\Omega)$. We obtain
\begin{eqnarray*}
&&\tau(u_0^{-})+\gamma\|u_0^{-}\|_2^2=0\ (\mbox{see}\ (\ref{eq7}))\\
&\Rightarrow& \hat{c}\|u_0^{-}\|^2\leq 0\ (\mbox{see}\ (\ref{eq4})),\ \mbox{hence}\ u_0\geq 0,\ u_0\neq 0.
\end{eqnarray*}
Then (\ref{eq26}) becomes
\begin{eqnarray}\label{eq27}
&& A(u_0)+\beta(z)u_0=N_f(u_0)-\lambda u_0^{q-1}\ (\mbox{see}\ (\ref{eq7}))\nonumber\\
&\Rightarrow& -\Delta u_0(z)+\beta(z)u_0(z)=f(z,u_0(z))-\lambda u_0(z)^{q-1}\ \mbox{a.e. in}\ \Omega,\qquad \frac{\partial u_0}{\partial n}=0\ \mbox{on}\ \partial\Omega.\\
&&(\mbox{see Motreanu and Papageorgiou\ \cite{6}})\nonumber.
\end{eqnarray}
Using Lemmata $5.1$ and $5.2$ of Wang \cite{14}, as in Papageorgiou and R\u adulescu \cite{10} (see the proof of Proposition $3.2$, p. $303$), we have that $u_0\in C_{+}\setminus\{0\}$. Harnack's inequality (see Pucci and Serrin \cite[p. 163]{13}), implies that
$$u_0(z)>0\ \mbox{for all}\ z\in\Omega.$$
\begin{claim}
$u_0$ is a local minimizer of $\varphi_{\lambda}$.
\end{claim}
According to Proposition \ref{p2}, it suffices to show that $u_0$ is a local $C^1(\overline{\Omega})$-minimizer of $\varphi_{\lambda}$. We argue by contradiction. So, suppose that $u_0$ is not a local $C^1(\overline{\Omega})$-minimizer of $\varphi_{\lambda}$. Then we can find $\{u_n\}_{n\geq 1}\subseteq C^1(\overline{\Omega})$ such that
$$u_n\to u_0\ \mbox{in}\ C^1(\overline{\Omega})\ \mbox{and}\ \varphi_{\lambda}(u_n)<\varphi_{\lambda}(u_0)\ \mbox{for all}\ n\geq 1.$$
From (\ref{eq7}) it is clear that $\varphi_{\lambda}\Big|_{C_{+}}=\hat{\varphi}_{\lambda}^{+}\Big|_{C_{+}}$. So, we have
\begin{eqnarray}\label{eq28}
0&>&\varphi_{\lambda}(u_n)-\varphi_{\lambda}(u_0)\nonumber\\
&=&\varphi_{\lambda}(u_n)-\hat{\varphi}_{\lambda}^{+}(u_0)\nonumber\\
&\geq& \varphi_{\lambda}(u_n)-\hat{\varphi}_{\lambda}^{+}(u_n)\ (\mbox{see}\ (\ref{eq25}))\nonumber\\
&=&\frac{1}{2}\tau(u_n)+\frac{\lambda}{q}\|u_n\|_q^q-\displaystyle\int_{\Omega}F(z,u_n)dz\nonumber\\
&&-\frac{1}{2}\tau(u_n)-\frac{\gamma}{2}\|u_n^{-}\|_2^2-\frac{\lambda}{q}\|u_n^{+}\|_q^q+\displaystyle\int_{\Omega}F(z,u_n^{+})dz\ (\mbox{see}\ (\ref{eq7}))\nonumber\\
&=&\frac{\lambda}{q}\|u_n^{-}\|_q^q-\frac{\gamma}{2}\|u_n^{-}\|_2^2-\displaystyle\int_{\Omega}F(z,-u_n^{-})dz\nonumber\\
&\geq&\frac{\lambda}{q}\|u_n^{-}\|_q^q-\frac{\gamma}{2}\|u_n^{-}\|_2^2-c_4\|u_n^{-}\|_2^2\ (\mbox{see}\ (\ref{eq21}))\nonumber\\
&\geq&\displaystyle\int_{\Omega}\left[\frac{\lambda}{q}-\left(\frac{\gamma}{2}+c_4\right)\|u_n^{-}\|_{\infty}^{2-q}\right](u_n^{-})^qdz.
\end{eqnarray}
Because $u_0(z)>0$ for all $z\in\Omega$ and $u_n\to u_0$ in $C^1(\overline{\Omega})$, it follows that
$$u_n^{-}\to 0\ \mbox{in}\ C(\overline{\Omega}).$$
So, from (\ref{eq28}) we see that we can find $n_0\in\NN$ such that
$$0>\displaystyle\int_{\Omega}\left[\frac{\lambda}{q}-\left(\frac{\gamma}{2}+c_4\right)\|u_n^{-}\|_{\infty}^{2-q}\right](u_n^{-})^qdz\geq 0\ \mbox{for all}\ n\geq n_0,$$
a contradiction. This proves the Claim.

From Proposition \ref{p7} we know that $u=0$ is a local minimizer of $\varphi_{\lambda}$, too. Then as in Aizicovici, Papageorgiou and Staicu \cite{1} (see the proof of Proposition $29$), we can find $\varrho\in (0,1)$ small such that
\begin{eqnarray}\label{eq29}
\hat{\varphi}_{\lambda}^{+}(u_0)<0=\hat{\varphi}_{\lambda}^{+}(0)<\inf\left[\hat{\varphi}_{\lambda}^{+}(u): \|u\|=\varrho\right]=\hat{m}_{\lambda,\varrho}^{+}.
\end{eqnarray}
Then (\ref{eq29}) and Corollary \ref{c5} permit the use of the mountain pass theorem. So, we can find $\hat{u}\in H^1(\Omega)$ such that
\begin{eqnarray}\label{eq30}
\hat{u}\in K_{\hat{\varphi}_{\lambda}^{+}}\ \mbox{and}\ 0=\hat{\varphi}_{\lambda}^{+}(0)<\hat{m}_{\lambda,\varrho}^{+}\leq \hat{\varphi}_{\lambda}^{+}(\hat{u}).
\end{eqnarray}
From (\ref{eq29}) and (\ref{eq30}) it follows that $\hat{u}\notin \{0,u_0\}$, $\hat{u}\in C_{+}\setminus\{0\}$ and solve $(P_{\lambda})$ (this is established as we did for $u_0$) and via Harnack's inequality we have $\hat{u}(z)>0$ for all $z\in\Omega$.

Similarly, working with $\hat{\varphi}_{\lambda}^{-}$, we produce $v_0,\hat{v}\in (-C_{+})\setminus\{0\}$ solutions of $(P_{\lambda})$ such that
$$v_0(z),\hat{v}(z)<0\ \mbox{for all}\ z\in\Omega$$
with $v_0$ local minimizer of $\varphi_{\lambda}$ and $0<\varphi_{\lambda}(\hat{v})$.
\end{proof}

\section{Five Solutions}\label{sec4}
To produce a fifth solution (but without any sign information), we need to restrict the behavior of the perturbation $f(z,\cdot)$ near the origin.

So, the new hypotheses on $f(z,x)$ are the following:

\smallskip
$H_2$: $f:\Omega\times\RR\to\RR$ is a Carath\'eodory function such that  $f(z,0)=0$ for a.a. $z\in\Omega$ and
\begin{itemize}
\item[(i)] $|f(z,x)|\leq a(z)(1+|x|)$ for a.a. $z\in\Omega$, all $x\in\RR$ with $a\in L^{\infty}(\Omega)_{+}$;
\item [(ii)] $\displaystyle\limsup_{x\to\pm\infty}\frac{f(z,x)}{x}\leq\hat{\lambda}_1(\beta)$ uniformly for a.a. $z\in\Omega$;
\item [(iii)] $\displaystyle\lim_{x\to +\infty}[f(z,x)x-2F(z,x)]=+\infty$ uniformly for a.a. $z\in\Omega$;
\item [(iv)] there exist an integer $m\geq 2$ and functions $\eta_0,\hat{\eta}_0\in L^{\infty}(\Omega)$ such that
\begin{eqnarray*}
&&\hat{\lambda}_m(\beta)\leq \eta_0(z)\leq\hat{\eta}_0(z)\leq\hat{\lambda}_{m+1}(\beta) \ \mbox{a.e. in}\ \Omega,\ \hat{\lambda}_m(\beta)\neq\eta_0,\ \hat{\lambda}_{m+1}(\beta)\neq\hat{\eta}_0,\\
&&\eta_0(z)\leq \displaystyle\liminf_{x\to 0}\frac{f(z,x)}{x}\leq\displaystyle\limsup_{x\to 0}\frac{f(z,x)}{x}\leq\hat{\eta}_0(z)\ \mbox{uniformly for a.a.}\ z\in\Omega,\\
&&F(z,x)\leq\frac{\hat{\lambda}_{m+1}(\beta)}{2}x^2\ \mbox{for a.a.}\ z\in\Omega,\ \mbox{all}\ x\in\RR.
\end{eqnarray*}
\end{itemize}

Let $Y=\oplus_{k=1}^{m}E(\hat{\lambda}_k(\beta))$ and $V=\overline{\oplus_{k\geq m+1}E(\hat{\lambda}_k(\beta))}$. We have the orthogonal direct sum decomposition
$$H^1(\Omega)=Y\oplus V.$$
We check the behavior of $\varphi_{\lambda}$ with respect to this decomposition.
\begin{prop}\label{p9}
If hypotheses $H_0$ and $H_2$ hold, then we can find $\hat{\lambda}^*\in (0,\lambda^*]$ such that  for all $\lambda\in (0,\hat{\lambda}^*)$ there exists $\varrho=\varrho(\lambda)>0$ for which we have
$$\varphi_{\lambda}\Big|_{\partial B_{\varrho}\cap Y}<0$$
(here $\lambda^*>0$ is as in Proposition \ref{p6} and $\partial B_{\varrho}=\{u\in H^1(\Omega): \|u\|=\varrho\}$).
\end{prop}
\begin{proof}
By virtue of hypotheses $H_2(i),(iv)$, given $\varepsilon>0$, we can find $c_6=c_6(\varepsilon)>0$ such that
\begin{eqnarray}\label{eq31}
F(z,x)\geq\frac{1}{2}(\eta_0(z)-\varepsilon)x^2-c_6|x|^r\ \mbox{for a.a.}\ z\in\Omega,\ \mbox{all}\ x\in\RR\ \mbox{with}\ r>2.
\end{eqnarray}
For all $u\in Y$, we have
\begin{eqnarray*}
\varphi_{\lambda}(u)&=&\frac{1}{2}\tau(u)+\frac{\lambda}{q}\|u\|_q^q-\displaystyle\int_{\Omega}F(z,u)dz\\
&\leq&\frac{1}{2}\left[\tau(u)-\displaystyle\int_{\Omega}\eta_0(z)u^2dz+\varepsilon\|u\|^2\right]+c_7\left(\|u\|^r+\frac{\lambda}{q}\|u\|^q\right)\\
&&(\mbox{for some}\ c_7>0\ (\mbox{see}\ (\ref{eq31})))\\
&\leq&\frac{1}{2}\left[-\hat{\xi}_0+\varepsilon\right]\|u\|^2+c_7\left(\|u\|^r+\frac{\lambda}{q}\|u\|^q\right)\ (\mbox{see Proposition}\ \ref{p3}).
\end{eqnarray*}
Choosing $\varepsilon\in (0,\hat{\xi}_0)$, we have
$$\varphi_{\lambda}(u)\leq \left[-c_8+c_7\left(\|u\|^{r-2}+\frac{\lambda}{q}\|u\|^{q-2}\right)\right]\|u\|^2\ \mbox{for some}\ c_8>0.$$
As in the proof of Proposition \ref{p6}, by considering the function $\mu_{\lambda}(\tau)=\tau^{r-2}+\frac{\lambda}{q}\tau^{q-2}$, we can find $\tilde{\lambda}^*>0$ such that  for all $\lambda\in (0,\tilde{\lambda}^*)$ there exists $\varrho=\varrho(\lambda)>0$ for which we have
$$\varphi_{\lambda}(u)<0\ \mbox{for all}\ u\in \partial B_{\varrho}\cap Y.$$
Finally let $\hat{\lambda}^*=\min\{\tilde{\lambda}^*,\lambda^*\}$.
\end{proof}

\begin{prop}\label{p10}
If hypotheses $H_0$ and $H_2$ hold, then $\varphi_{\lambda}\Big|_{V}\geq 0$.
\end{prop}
\begin{proof}
Using hypothesis $H_2(iv)$, for every $u\in V$, we have
$$\varphi_{\lambda}(u)\geq \frac{1}{2}\tau(u)-\frac{\hat{\lambda}_{m+1}(\beta)}{2}\|u\|^2\geq 0.$$
\end{proof}
Now we can produce a fifth nontrivial solution.
\begin{prop}\label{p11}
If hypotheses $H_0$ and $H_2$ hold and $\lambda\in (0,\hat{\lambda}^*)$, then problem $(P_{\lambda})$ has a fifth nontrivial solution $y_0\in C^1(\overline{\Omega})$.
\end{prop}
\begin{proof}
Proposition \ref{p4} implies that the energy functional $\varphi_{\lambda}$ is bounded below. This fact together with Propositions \ref{p9} and \ref{p10} permit the use of Theorem $3.1$ of Perera \cite{12}. So, we can find $y_0\in H^1(\Omega)$ such that
\begin{eqnarray}\label{eq32}
&&y_0\in K_{\varphi_{\lambda}},\ \varphi_{\lambda}(y_0)<0=\varphi_{\lambda}(0)\ \mbox{and}\ C_{d_m-1}(\varphi_{\lambda},0)\neq 0\ (d_m=\dim \oplus_{k=1}^m E(\hat{\lambda}_k(\beta))\geq 2).
\end{eqnarray}
Since $\varphi_{\lambda}(\hat{u}), \varphi_{\lambda}(\hat{v})>0$ (see Proposition \ref{p8}), we have that $y_0\notin\{a,\hat{u},\hat{v}\}$ (see also (\ref{eq32})). From Proposition \ref{p8} we know that $u_0,v_0$ are local minimizers of $\varphi_{\lambda}$. Hence
\begin{eqnarray}\label{eq33}
&&C_k(\varphi_{\lambda},u_0)=C_k(\varphi_{\lambda},v_0)=\delta_{k,0}\ZZ\ \mbox{for all}\ k\geq 0.
\end{eqnarray}
Comparing (\ref{eq32}) and (\ref{eq33}), we see that $y_0\notin\{u_0,v_0\}$. Therefore $y_0$ is the fifth nontrivial solution of $(P_{\lambda})$ (since $y_0\in K_{\varphi_{\lambda}}$, see (\ref{eq32})) and as before using the regularity result of Wang \cite{14}, we have $y_0\in C^1(\overline{\Omega})$.
\end{proof}

So, we can state the following multiplicity theorem for problem $(P_{\lambda})$.
\begin{theorem}\label{th12}
If hypotheses $H_0$ and $H_2$ hold, then there exists $\hat{\lambda}^*>0$ such that  for all $\lambda\in (0,\hat{\lambda}^*)$ problem $(P_{\lambda})$ has at least five nontrivial solutions
\begin{eqnarray*}
&&u_0,\hat{u}\in C_{+},\ u_0(z),\hat{u}(z)>0\ \mbox{for all}\ z\in\Omega,\\
&&v_0,\hat{v}\in -C_{+},\ v_0(z),\hat{v}(z)<0\ \mbox{for all}\ z\in\Omega,\\
&&y_0\in C^1(\overline{\Omega})\setminus\{0\},
\end{eqnarray*}
with $u_0,v_0$ local minimizers of the energy functional and
$$\varphi_{\lambda}(u_0),\ \varphi_{\lambda}(v_0),\ \varphi_{\lambda}(y_0)<0=\varphi_{\lambda}(0)<\varphi_{\lambda}(\hat{u}),\ \varphi_{\lambda}(\hat{v}).$$
\end{theorem}

\begin{remark}
It is an interesting open question whether the fifth solution $y_0$ is nodal (that is sign changing).
\end{remark}




\end{document}

\begin{theorem}\label{th1}
Assume that $\varphi\in C^1(X)$ satisfies the $PS$-condition, $u_0, u_1\in X$, with $\|u_1-u_0\|>r>0$,
$$\max\{\varphi(u_0),\varphi(u_1)\}<\inf\left[\varphi(u): \|u-u_0\|=\varrho\right]=m_{\varrho}$$
and $c=\inf_{\gamma\in\Gamma}\max_{t\in [0,1]}\varphi(\gamma(t))$, where $\Gamma=\{\gamma\in C([0,1],X): \gamma(0)=u_0,\ \gamma(1)=u_1\}.$

Then $c\geq m_{\varrho}$ and $c$ is a critical value of $\varphi$.
\end{theorem} 